Author's response to reviews

Title: Polymorphous low-grade adenocarcinoma arising in tongue

Authors:

Ruchi Gupta SR (ruchipgi@yahoo.co.in)
Kirti Gupta AP (kirtigupta10@yahoo.co.in)
Rijuneeta Gupta AP (rijuneeta@yahoo.com)

Version: 4 Date: 24 March 2009

Author's response to reviews: see over
To,
The Editor-in-chief,
JMCR

Dear Sir,

We thank you for the peer review of the article titled “Polymorphous low-grade adenocarcinoma arising in tongue” by Gupta et al for publication in your journal as “Case report”.

We have revised the manuscript as per the very valuable comments and suggestions made by the reviewers. The point-wise rebuttal is also uploaded. I hope that the revised manuscript is suitable for publication in your esteemed journal.

The formatting matters that required your attention before our manuscript can be considered for peer review, have been taken care of. We deeply regret the delay in submitting the revised manuscript.

Thankyou.

Hope to hear from you.

Thanking you,
Warm Regards,

Dr. Kirti Gupta
Asstt. Professor,
Dept of Histopathology,
PGIMER, Chandigarh.
India.